Author's response to reviews

Title: Prognostic value of preoperative neutrophil-to-lymphocyte and platelet-to-lymphocyte ratios, and multiphasic renal tomography findings in histological subtypes of renal cell carcinoma

Authors:

Suat Keskin (drsuatkeskin@yahoo.com)
Zeynep Keskin (zkeskin@ymail.com)
Hakan Hakki Taskapu (drtaskapu@hotmail.com)
Havva Kalkan (drhavyaradyoloji2011@gmail.com)
Mehmet Kaynar (mekaynar@gmail.com)
Necdet Poyraz (necdetpoyraz@gmail.com)
Hatice Toy (haticetoy@hotmail.com)

Version: 6
Date: 5 November 2014

Author's response to reviews:

Dear editor,

1) I wrote this significant results as "In two-way analysis of variance test, statistically significant results which were influenced by mortality (p=0.028) and were found between renal cell carcinoma subtypes in the computed tomography density of corticomedullary phase (p=0.001)." in line 39-42 and as "In two-way analysis of variance test, statistically significant results which were influenced by mortality (p=0.028) and were found between RCCs subtypes in the CT density of corticomedullary phase (p=0.001)." in line 188-190.

2) I added "were enrolled in the study after providing written informed consent." in line 74-75.

3) I changed to pathological grading as "for pathological grading " in line 85.

4) I changed Levene as "Levene's" in line 128.

5) I changed nephrogenic as "nephrographic" in line 183, 187, 188, 234, 242.

6) I changed sarcomatiod as "sarcomatoid" in line 241.

7) I changed chromofobe as "chromophobe" in table 1 and 2.

8) I changed creatinin as "creatinine" in table 3.

Sincerely,

Dr. Suat Keskin